Re: Manuscript 71644; “Glomangiomatosis: Case report with immunohistochemical study and literature review”

Dear Dr. Wang,

Thank you very much for your decision letter and advice on our manuscript. We also thank the reviewers for the constructive comments and suggestions. We have revised the manuscript accordingly, and all amendments are indicated by red font in the revised manuscript. In addition, our point-by-point responses to the reviewers’ comments are listed below this letter.

This revised manuscript has been edited and proofread by Medjaden Inc.

We hope that our revised manuscript is now acceptable for publication in World Journal of Clinical Cases and look forward to hearing from you soon.

With best wishes,
Yours sincerely,
Phone: 18505387050

Email: zhangshp01@163.com
Replies to Reviewer 1

Specific Comments

1. LANGUAGE POLISHING REQUIREMENTS FOR REVISED MANUSCRIPTS SUBMITTED BY AUTHORS WHO ARE NON-NATIVE SPEAKERS OF ENGLISH

Response: Thank you for your insightful suggestion. The article was edited and proofread by a professional native English medical editor at Medjaden Bioscience.

Replies to editorial office’s comments

Replies to science editor

1. The choice of the references is outdated, the author should supplement the latest literature.

Response: The references have been revised according to your suggestion.

2. Supplement the biochemical examination results of patients.

Response: Biochemical examination results of patients have been supplemented

Replies to company editor-in-chief. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. Authors are required to provide standard three-line tables, that is, only the top line, bottom line, and column line are displayed, while other table lines are hidden. The contents of each cell in the table should conform to the editing specifications, and the lines of each row or column of the table should be aligned. Do not use carriage returns or spaces to replace lines or vertical lines and do not segment cell content. Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

Response: The figure has been put in the PPT, and the form has been put in the document separately. The content and format of the article were modified in accordance with the requirements of your journal.